Author’s response to reviews

Title: Unique electrocardiographic pattern “w” wave in lead I of idiopathic ventricular arrhythmias arising from the distal great cardiac vein

Authors:
Teng Li (gdmc99@163.com)
Qiong Xu (47139772@qq.com)
Xian-zhang Zhan (gdzxz@163.com)
Yu-mei Xue (xymgdeci@163.com)
Hong-tao Liao (hometown@163.com)
Yi-fu Li (yeefy@163.com)
Konstantinos Letsas (k.letsas@gmail.com)
Shu-lin Wu (doctorwushuling@163.com)

Version: 3 Date: 14 Mar 2019

Author’s response to reviews:

Dear Editor:

We are truly grateful to you for all critical comments. Based on these comments and suggestions, we have made careful modifications to the last manuscript.

1. Thanks. In Declarations section, line 301, page 10, "Not applicable" had been changed to "The datasets used and/or analysed during the current study available from the corresponding author on reasonable request." for the availability of data and materials section.

2. Manuscript had been uploaded as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files also were clean versions. Figures (and additional files) also were uploaded as separate files.